Author’s response to reviews

Title: Translational research on reserve against neurodegenerative disease: Consensus report of the International Conference on Cognitive Reserve in the Dementias and the Alzheimer’s Association Reserve, Resilience and Protective Factors Professional Interest Area working groups

Authors:

Robert Perneczky (robert.perneczky@med.uni-muenchen.de)
Gerd Kempermann (gerd.kempermann@dzne.de)
Amos Korczyn (amoskor@tau.ac.il)
Fiona Matthews (fiona.matthews@newcastle.ac.uk)
M Arfan Ikram (m.a.ikram@erasmusmc.nl)
Nikolaos Scarmeas (ns257@columbia.edu)
Gael Chetelat (chetelat@cyceron.fr)
Yaakov Stern (ys11@cumc.columbia.edu)
Michael Ewers (michael.ewers@med.uni-muenchen.de)

Version: 1 Date: 11 Sep 2018

Author’s response to reviews:

Dear Dr Graham,

Thank you for bringing to my attention that certain parts of our manuscript “Translational research on reserve against neurodegenerative disease” required rewriting to meet the high standards of BMC Medicine for an original contribution. The article was amended accordingly, and I am resubmitting it for publication as an opinion paper.

Yours sincerely,

Robert Perneczky